Author’s response to reviews

Title: Differential modulation of nicotine-induced gemcitabine resistance by GABA receptor agonists in pancreatic cancer xenografts and in vitro

Authors:

Jheelam Banerjee (jbanerje@utk.edu)
Hussein A.N. Al-Wadei (halwadei@utk.edu)
Mohammed H. Al-Wadei (malwadei@utk.edu)
Koami Dagnon (dagnon.clement@gmail.com)
Hildegard M. Schuller (hmsch@utk.edu)

Version: 2 Date: 21 April 2014

Author’s response to reviews:

In accord with your request, page numbers and line numbering have been added to the manuscript.